Author's response to reviews

Title: Body composition among Sri Lankan infants by 18O dilution method and the validity of anthropometric equations to predict body fat against 18O dilution

Authors:

Thushari Bandara (wvthush@yahoo.com)
Manjula Hettiarachchi (manjula.hettiarachchi@gmail.com)
Chandrani Liyanage (kdceliyanage@gmail.com)
Sujeewa Amarasena (sujeewaamarasena@yahoo.com)
William Wai-Lun Wong (wwong@bcm.edu)

Version: 2
Date: 2 December 2014

Author's response to reviews:

Dear Editor

BMC Pediatrics,

Thank you so much for the advice given on revising of our manuscript.

There were three things to do. Firstly to add the line numbers and page numbers to the document, secondly to include the tables in the main document itself and format the tables and finally to include a section for 'competing interest'.

I have done all of the above modifications to the manuscript.

Kindly let me know if there is anything else required for us to do.

Thank you for your kind cooperation.

Sincerely

Thushari Bandara
Corresponding author